Author’s response to reviews

Title: Diversification of U.S. Medical Schools via Affirmative Action Implementation

Authors:

Shaheen E Lakhan (slakhan@fas.harvard.edu)

Version: 5 Date: 24 Aug 2003

August 23, 2003

Emma Veitech, BioMed Central
Janine C Edwards, Reviewer
Stephen A Magnus, Reviewer

Dear Emma,

Thank you again for the fast and thorough review process. I am submitting a revision of the "Diversification of U.S. Medical School via Affirmative Action Implementation" manuscript. I have edited my manuscript to address the advice of the reviewers. I have made typographical/grammatical edits and the errors produced by format transfer have been resolved. The 1910 Flexner Report and associated consequences on women and minority medical schools have been merged into the Background to provide a more accurate account of the medical schools of that era, at the suggestion of Stephen Magnus. The Discussion section appears at the appropriate location. Also, the organization of the paper has been modified and strengthened, and the "roadmap" altered to "drive" the reader in a more smooth fashion.

Thank you for your editing and reviewing assistance.

Sincerely,

Shaheen Emmanuel Lakhan

Harvard Institutes of Medicine
4 Blackfan Circle, Room 208
Boston, MA 02115-5713
Tel/Fax: 909-498-8657
Email: SLakhan@fas.Harvard.edu